Author's response to reviews

Title: Cross-cultural validation of the Educational Needs Assessment Tool in RA in 7 European countries

Authors:

Mwidimi Ndosi (m.e.ndosi@leeds.ac.uk)
Alan Tennant (a.tennant@leeds.ac.uk)
Ulrika Bergsten (ulrika.bergsten@spenshult.se)
Marja Leena Kukkurainen (marja.kukkurainen@phnet.fi)
Pedro Machado (pedrommcmachado@gmail.com)
Jenny de la Torre-Aboki (delatorre_jen@gva.es)
Thea P. M. Viet Vieland (t.p.m.vliet_vlieland@lumc.nl)
Heidi A. Zangi (heidi.zangi@diakonsyk.no)
Jackie Hill (j.hill@leeds.ac.uk)

Version: 5 Date: 23 May 2011

Author's response to reviews: see over
The Editor,
BMC Musculoskeletal Disorders
BioMed Central Ltd

Dear Editor

**Re: Cross-cultural validation of the Educational Needs Assessment Tool in RA in 7 European countries. MS: 1480950561481438**

Thank you for your feedback on the above manuscripts. We are pleased to re-submit the above article for publication in the BMC Musculoskeletal Disorders having done the formatting changes advised by the Production Team.

1. Table numbers have been corrected.

Sincerely yours,

Mwidimi Ndosi (Corresponding author)
Academic & Clinical Unit for Musculoskeletal Nursing (ACUMeN)
Section of Musculoskeletal Disease
Leeds Institute of Molecular Medicine, University of Leeds
2nd Floor, Chapel Allerton Hospital
Chapeltown Road
Leeds, LS7 4SA
United Kingdom.
Tel +441133924859, Fax +441133924755
Email: m.e.nndosi@leeds.ac.uk